\documentclass[12pt,letterpaper]{article}
\usepackage[letterpaper,margin=1.0in]{geometry}
\usepackage{caption}
\usepackage{subcaption}
%\usepackage{graphicx,amsmath,bm,floatflt,multirow,epstopdf}
\usepackage{graphicx,amsmath,bm,multirow,epstopdf}
\usepackage{wrapfig}
\usepackage[utf8]{inputenc}
\usepackage{cite}
\usepackage{url}
\usepackage[toc]{appendix}
\usepackage{bibunits}

\usepackage{etoolbox}

\pagestyle{plain}
\bibliographystyle{aip}
\defaultbibliographystyle{aip}

\begin{document}

%Here is a reference \cite{Anisovich:2018yoo,Kunkel:2017src,Roy:2017qwv,Akbar:2017uuk,                     
%Amaryan:2017ldw,Anisovich:2017pox,Collins:2017vev,Mattione:2017fxc,             
%Anisovich:2017rpe,Dugger:2017zoq,Collins:2017sgu,AlGhoul:2017nbp,               
%Paterson:2016vmc,Ghoul:2015ifw}
%\bibliography{bibtexrefs2018}{}

\input MikeBioSketch.tex 

%Blablabla said Nobody ~\cite{Nobody06}.
%\bibliography{tmpTest.bib}{}


\end{document}
